Author's response to reviews

Title: Contribution of main causes of death to social inequalities in mortality in the whole population of Scania, Sweden

Authors:

Maria Rosvall (Maria.rosvall@med.lu.se)
Basile Chaix (Basile.chaix@med.lu.se)
John Lynch (John.lynch@mcgill.ca)
Martin Lindstrom (Martin.lindstrom@med.lu.se)
Juan Merlo (Juan.merlo@med.lu.se)

Version: 8 Date: 24 March 2006

Author's response to reviews:

Page 2- The secondary title page has been removed.

The 4 broad tables are resubmitted as additional files 1-4.